Author’s response to reviews

Title: Influenza Vaccination among children with Idiopathic Nephrotic Syndrome: an investigation of practices

Authors:

Roman Klifa (klifaroman@msn.com)
Julie Toubiana (julie.toubiana@aphp.fr)
Alizée Michel (alizee.michel@aphp.fr)
Nathalie Biebuyck (nathalie.biebuyck@aphp.fr)
Marina Charbit (marina.charbit@aphp.fr)
Laurence Heidet (laurence.heidet@aphp.fr)
Saoussen Krid (saoussen.krid@aphp.fr)
Pauline Krug (pauline.krug@aphp.fr)
Rémi Salomon (remi.salomon@aphp.fr)
Olivia Boyer (olivia.boyer@aphp.fr)

Version: 2 Date: 26 Jan 2019

Author’s response to reviews:

We would like to thank you for giving us the opportunity to resubmit a revised version of our manuscript entitled “Influenza vaccination among children with idiopathic nephrotic syndrome: an investigation of practices”. The comments made by the reviewers have been taken into account, and the manuscript has been modified accordingly. The revised version has been seen and approved by all co-authors. We hope you will find this revised version suitable for publication in the BMC Nephrology, and we would be pleased to provide you with any additional information you may require.

We look forward to hearing from you,

Sincerely,